Author’s response to reviews

Title: Screening differentially expressed genes of pancreatic cancer between Mongolian and Han people using bioinformatics technology

Authors:

Jiasheng Xu (1821286450@qq.com)
Kaili Liao (1192784983@qq.com)
ZHONGHUA FU (fzhncuedu@163.com)
ZHENFANG XIONG (zhenfangxiongedu@163.com)

Version: 4 Date: 21 Feb 2020

Author’s response to reviews:

First of all, thank you very much for your support and guidance to our article! We have revised it in accordance with your requirements. We have answered your questions one by one. We hope you will be satisfied with it, thank you!

Refutation and question reply to the editor:

Question 1: Please help me understand what you are referring as 'race concerns' that prevent you from adding the data to your submission.

Reply to question 1:

Thank you very much for your question! Since the previous requirements were to provide sequencing results of normal tissues of the Mongolian and Han nationalities, we were concerned that genetic differences before different races would be involved, so we failed to provide this part of the data. We have now communicated with the patients and obtained their consent to publish this part of the data. Therefore, we have now added that part of the data to the article. Thank you for your understanding and concern! We hope our response will make you satisfied, thank you!

Question 2: please provide the additional data attached in your revision.
Reply to question 2:

Thank you very much for your suggestion! We have supplemented the additional data attached in our revision according to your suggestions. We hope that our changes and responses will make you satisfied, thank you!

Refutation and question reply to reviewer 4:

Question 1: It is totally a valid scientific question to know whether the deferentially expressed genes between Mongolian and Han PDACs (Figure 6) are reflected in adjacent "normal" tissues. These data need to be provided.

Reply to question 1:

Thank you very much for your suggestions! We have now communicated with the patients and obtained their consent to publish this part of the data. We have supplemented the additional data and redrawn Figure 6 according to your suggestions. See specific changes in the article. We really appreciate your suggestions, which will help us improve the quality of the article! We really appreciate your suggestions, which will help us improve the quality of the article! We hope that our modification and responses will make you satisfied, thank you!